Author's response to reviews

Title: Down-regulation of SFRP1 as a putative tumor suppressor gene can contribute to human hepatocellular carcinoma

Authors:

Jian Huang (huangj@chgc.sh.cn)
Yun-li Zhang (zhangyl@chgc.sh.cn)
Xiao-mei Teng (tengxm@chgc.sh.cn)
Yun Lin (liny@chgc.sh.cn)
Da-li Zheng (zhengdl@chgc.sh.cn)
Pengyuan Yang (yangpy@fudan.edu.cn)
Ze-guang Han (zghan64@hotmail.com)

Version: 5 Date: 11 July 2007

Author's response to reviews: see over
Dear Editor,

Enclosed herewith for your consideration is the revised version of our manuscript (MS: 2083078131130252) entitled “Down-regulation of SFRP1 as a putative tumor suppressor gene can contribute to human hepatocellular carcinoma” by Jian Huang, Yun-Li Zhang, Xiao-Mei Teng, Yun Lin, Haihui Sheng, Da-Li Zheng, Pengyuan Yang, Ze-Guang Han.

We have revised the format of our manuscript according to your requirements:

1. We removed the Key words from the title page.
2. We changed ‘Conclusion’ to ‘Conclusions’.
3. We changed ‘Figure Legends’ to ‘Figure legends’.
4. We finished a final check of our manuscript and revised some errors.

We hope that our manuscript is now available for publication. Please let me know if we can provide further information.

Sincerely yours,

Ze-Guang Han PhD, MD
Chinese National Human Genome Center at Shanghai
351, Guo Shou-Jing Road
Shanghai 201203,
China
Tel: 86-21-50801325
FAX: 86-21-50800402
Email: hanzg@chgc.sh.cn